Author’s response to reviews

Title: Effect of sedation with dexmedetomidine or propofol on gastrointestinal motility in lipopolysaccharide-induced endotoxemic mice

Authors:

Haiqing Chang (haiqingchang123@163.com)
Shuang Li (lish09doctor@sina.com)
Yansong Li (liyansong.inter@163.com)
Hao Hu (huhao@mail.xjtu.edu.cn)
Bo Cheng (chengboclh@163.com)
Jiwen Miao (doctormiaojw@163.com)
Hui Gao (imevm@163.com)
Hongli Ma (tahongli@126.com)
Yanfeng Gao (gaoyf2009@126.com)
Qiang Wang (dr.wangqiang@139.com)

Version: 1 Date: 19 May 2020

Author’s response to reviews:

Dear editor,
we appreciate for Editors’ warm work earnestly. We have revised the manuscript according to the editor’s requirement. In detail, we clarified sacrifice methods in paragraph 1 of methods section. We would be glad to respond to any further questions and comments that you may have. And we hope that the revision is acceptable and look forward to hearing from you soon.

Sincerely,
Qiang Wang